

Biology, Chemistry, and Environmental Sciences Faculty Articles and Research Science and Technology Faculty Articles and Research

11-3-2023

Volcanic Diffuse Volatile Emissions Tracked by Plant Responses Detectable From Space

Robert R. Bogue *McGill University*

Peter M. J. Douglas *McGill University*

Joshua B. Fisher Chapman University, jbfisher@chapman.edu

John Stix McGill University

Follow this and additional works at: https://digitalcommons.chapman.edu/sees_articles

Commons, Other Plant Sciences Commons, and the Volcanology Commons

Recommended Citation

Bogue, R. R., Douglas, P. M. J., Fisher, J. B., & Stix, J. (2023). Volcanic diffuse volatile emissions tracked by plant responses detectable from space. *Geochemistry, Geophysics, Geosystems, 24*, e2023GC010938. https://doi.org/10.1029/2023GC010938

This Article is brought to you for free and open access by the Science and Technology Faculty Articles and Research at Chapman University Digital Commons. It has been accepted for inclusion in Biology, Chemistry, and Environmental Sciences Faculty Articles and Research by an authorized administrator of Chapman University Digital Commons. For more information, please contact laughtin@chapman.edu.

Volcanic Diffuse Volatile Emissions Tracked by Plant Responses Detectable From Space

Comments

This article was originally published in *Geochemistry, Geophysics, Geosystems*, volume 24, in 2024. https://doi.org/10.1029/2023GC010938

Creative Commons License



This work is licensed under a Creative Commons Attribution 4.0 License.

Copyright The authors

Geochemistry, Geophysics, Geosystems[°]

RESEARCH ARTICLE 10.1029/2023GC010938

Key Points:

- Plant responses to elevated CO₂ may be a viable proxy for volcanic CO₂
- We have detected variations in diffuse volatile emissions in Yellowstone using plant responses
- Plants responded to changes in volatile emissions several years before hydrothermal activity was previously thought to have started

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

R. R. Bogue, robert.r.bogue@gmail.com

Citation:

Bogue, R. R., Douglas, P. M. J., Fisher, J. B., & Stix, J. (2023). Volcanic diffuse volatile emissions tracked by plant responses detectable from space. *Geochemistry, Geophysics, Geosystems*, 24, e2023GC010938. https://doi. org/10.1029/2023GC010938

Received 7 MAR 2023 Accepted 3 AUG 2023

Author Contributions:

Conceptualization: Robert R. Bogue, Joshua B. Fisher Investigation: Robert R. Bogue Methodology: Robert R. Bogue Supervision: Peter M. J. Douglas, John Stix Visualization: Robert R. Bogue Writing – original draft: Robert R. Bogue

Writing – review & editing: Peter M. J. Douglas, Joshua B. Fisher, John Stix

© 2023 The Authors. Geochemistry, Geophysics, Geosystems published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Volcanic Diffuse Volatile Emissions Tracked by Plant Responses Detectable From Space

Robert R. Bogue¹, Peter M. J. Douglas¹, Joshua B. Fisher², and John Stix¹

¹Department of Earth and Planetary Sciences, McGill University, Montreal, QC, Canada, ²Schmid College of Science and Technology, Chapman University, Orange, CA, USA

Abstract Volcanic volatile emissions provide information about volcanic unrest but are difficult to detect with satellites. Volcanic degassing affects plants by elevating local CO₂ and H₂O concentrations, which may increase photosynthesis. Satellites can detect plant health, or a reaction to photosynthesis, through a Normalized Difference Vegetation Index (NDVI). This can act as a potential proxy for detecting changes in volcanic volatile emissions from space. We tested this method by analyzing 185 Landsat 5 and 8 images of the Tern Lake thermal area (TLTA) in northeast Yellowstone caldera from 1984 to 2022. We compared the NDVI values of the thermal area with those of similar nearby forests that were unaffected by hydrothermal activity to determine how hydrothermal activity impacted the vegetation. From 1984 to 2000, plant health in the TLTA steadily increased relative to the background forests, suggesting that vegetation in the TLTA was fertilized by volcanic CO₂ and/or magmatic water. Hydrothermal activity began to stress plants in 2002, and by 2006, large swathes of trees were dying in the hydrothermal area. Throughout most of the 1990s, the least healthy plants were located in the area which became the epicenter of hydrothermal activity in 2000. These findings suggest that plant-focused measurements are sensitive to minor levels of volcanic unrest which may not be detected by other remote sensing methods, such as infrared temperature measurements. This method could be a safe and effective new tool for detecting changes in volatile emissions in volcanic environments which are dangerous or difficult to access.

Plain Language Summary Scientists often measure volatile emissions from volcanoes to understand how the magma underneath the volcano is behaving to anticipate potential volcanic hazards. These emissions are difficult and often hazardous to measure on the ground; therefore, measuring them with satellites would facilitate consistent and safe detection of changes in volcanic activity. It is nearly impossible to directly measure volcanic carbon dioxide and H_2O from space, so we need another method. We use plants as a proxy because they use carbon dioxide and water for photosynthesis, so if plants receive these extra gases from the volcano, it should improve plant health. We detected signals of this increased health from 1984 to 2001 in Yellowstone National Park, USA. One area of the forest that was exposed to volcanic gases was healthier than nearly identical, nearby forests growing without the influence of the volcano. This supports the idea that extra volcanic carbon dioxide and water promote tree health. As volcanic activity intensifies, it can harm plants. We detected increases in plant stress caused by increases in soil temperature and sulfur emissions before they were detectable by other types of satellites. Combining these two contrasting effects represents a promising new path for additional monitoring of active volcanoes.

1. Introduction

Carbon dioxide emissions are frequently one of the earliest precursors to volcanic eruptions but often are not detected due to the difficulty of differentiating them from atmospheric CO_2 . As magma ascends through the crust, CO_2 becomes increasingly insoluble and exsolves from the magma (Bottinga & Javoy, 1990; Gardner et al., 2000; Proussevitch & Sahagian, 1998; Sparks et al., 2018). These exsolved fluids quickly ascend through the faults and fracture systems in the crust and are emitted at the surface often before the rising magma (Nuccio & Paonita, 2001). Volcanic CO_2 emissions can be measured in situ with a variety of instruments, but this approach is time consuming, and measurements tend to be spatially and temporally limited. Many volcanoes are also relatively inaccessible and can be covered in thick vegetation and challenging topography, making routine measurements difficult. The National Academy of Sciences has recognized that remote sensing could overcome these difficulties and recently identified satellite detection of passive volcanic CO_2 degassing as one of the grand challenges of volcano sciences (Committee on Improving Understanding of Volcanic Eruptions et al., 2017).



Schwandner et al. (2017) successfully measured volcanic CO_2 emissions with the OCO-2 satellite, but could only detect a few of the highest degassing volcanoes in the world. Surface-level CO_2 anomalies need to be strongly anomalous to make a detectable impact on a satellite-based atmospheric column measurement of CO_2 due to the high concentrations of atmospheric CO_2 (Burton et al., 2013; Schwandner et al., 2017). This constraint inherently limits this technique to a small number of highly active volcanoes and highlights the need for an alternative way to measure volcanic CO_2 emissions from space. Other satellites such as the GHGSat Claire can detect CO_2 at higher spatial resolutions (Jervis et al., 2021) but require focused targets for observation, limiting their use for consistent monitoring of the ~1,400 active subaerial volcanoes worldwide.

Plant-focused measurements may offer a viable indicator for the spaceborne detection of volcanic volatile emissions. Determining how plants respond to rising levels of atmospheric CO₂ is also one of the most pressing questions in climate science and ecology (Cox et al., 2013; Friedlingstein et al., 2014). Measuring how plants respond to volcanic volatile emissions could provide crucial data for this effort as well as an effective way to detect degassing at hundreds of active but poorly monitored volcanoes around the world (Cawse-Nicholson et al., 2018; Friedlingstein et al., 2014). An exploratory study at Turrialba volcano in Costa Rica found a correlation between soil CO₂ fluxes and heavier δ^{13} C values in wood from nearby trees, which suggests that those trees incorporated significant amounts of volcanic CO₂ (Bogue et al., 2018). Tree ring samples from trees downwind of the volcanic plume from Turrialba had shifts in sulfur and carbon isotopes, which is caused by the incorporation of volcanic sulfur and sulfur-induced leaf function impairment, representing an increase in volcanic degassing (D'Arcy et al., 2019). Major Normalized Difference Vegetation Index (NDVI) anomalies preceded flank eruptions at Mt. Etna in Italy and Nyiragongo in the Democratic Republic of the Congo in the early 2000s, demonstrating that plants respond to and can potentially anticipate the location of fissure eruptions (Houlié et al., 2006). Two follow-up studies found that the trees on Mt. Etna are relatively insensitive to changes in temperature or water availability, suggesting that CO₂ fertilization may have contributed to the NDVI anomalies there (Seiler et al., 2017, 2021). Tree ring samples taken from nearby trees not destroyed in the eruption displayed significant ¹⁴C and δ^{18} O anomalies in the years leading up to the eruption, suggesting an uptake of volcanic CO₂ and H₂O (Seiler et al., 2017, 2021).

Water, sulfur dioxide, and hydrogen sulfide are the other primary gas species emitted by volcanic systems. Water is the dominant component of volcanic gas emissions, often making up more than 90% of the total gas released (Fischer, 2008; Shinohara, 2008). Similar to CO₂, it is difficult to measure with remote sensing methods due to the high concentrations of H₂O in the atmosphere (Girona et al., 2015). Elevated H₂O degassing likely preceded eruptions at Sabancaya, Mt. Etna, and Nyiragongo, establishing it as a potential target for volcano monitoring operations (Kern et al., 2017; Seiler et al., 2021). Volcanic H₂O degassing has typically been measured at fumaroles or volcanic vents; and we are unaware of any studies which focus on diffuse H₂O degassing. Unlike H₂O, volcanic SO₂ emissions are relatively easy to measure with satellites, and estimates of SO₂ degassing are now available for 106 volcanoes around the world (Carn et al., 2017). At volcanoes with well-developed hydrothermal systems, large quantities of sulfur are scrubbed by dissolution in the hydrothermal fluids while ascending through the crust, leading to areas with nearly pure CO, emissions at the surface (Kern et al., 2022; Symonds et al., 2001). Trees can respond to elevated concentrations of CO_2 in a variety of ways, including increasing photosynthesis and specific leaf area (Ainsworth & Long, 2005; Saban et al., 2019). NDVI is a measurement of chlorophyll abundance in a given area, which is a proxy for photosynthesis and tree health and thus has potential for detecting tree responses to changes in local volatile concentrations (Gamon et al., 1995). Volcanic activity can also negatively affect or kill plants, which decreases the NDVI of the area. Sulfur-bearing gas emissions, extremely high CO₂ concentrations, and/or increases in soil temperature can kill large areas of plants (Farrar et al., 1995; Jenkins et al., 2012). All three of these phenomena are typically associated with hydrothermal activity or shallow magma, whereas CO₂ emissions without accompanying sulfur gases or major soil temperature anomalies are usually indicative of the degassing of deeper pockets of magma (Symonds et al., 2001).

Large-scale ground temperature increases have recently been detected via remote sensing at a number of volcanoes prior to eruptions, including those which did not appear to produce any other early warning signs (Girona et al., 2021). Diffuse degassing is often strongly correlated with heat flow at volcanoes and hydrothermal systems, and these temperature anomalies are potentially associated with increased degassing (Chiodini et al., 2005; Finizola et al., 2006; Revil et al., 2011; Zimmer & Erzinger, 2003). Increased CO₂ emissions were detected prior to some of these eruptions, including the 2014 eruption of Pico de Fogo in Cape Verde and the 2009 eruption of Mt. Redoubt in Alaska (Bull et al., 2011; Global Volcanism Program, 2014), strengthening the link between diffuse degassing, diffuse heat flux, and eruptions.

In this study, we investigate the history of the Tern Lake thermal area (TLTA), a recently recognized active hydrothermal area in Yellowstone caldera in Wyoming, USA. Hydrothermal activity was identified from satellite images which captured visible hydrothermal alteration in the soil and a \sim 33,000 m² area of dead trees accompanied by temperature anomalies detected by thermal infrared imaging (Vaughan et al., 2020). These anomalies were first detected in satellite images from 2000, and from 2006 to 2015, the hydrothermally altered area expanded significantly (Vaughan et al., 2020). The principal goal of this study is to investigate how the trees in this area responded to changes in hydrothermal activity, and to determine if these responses provide additional information about the volcanic activity that thermal infrared and visible light images did not capture.

2. Materials and Methods

2.1. Study Site Description

Yellowstone is a large, restless caldera system characterized by a highly active hydrothermal system which emits an estimated 45 kt d⁻¹ of CO₂, 96% of which emanates from acid-sulfate hydrothermal areas (Werner & Brantley, 2003). Tern Lake is located in the northeast sector of Yellowstone just inside the caldera rim in an area which hosts a number of hydrothermal zones (Christiansen, 2001; Vaughan et al., 2020). Activity at the Tern Lake hydrothermal area was only recently recognized from satellite imagery that showed hydrothermally altered soil, dead trees, and infrared temperature anomalies (Vaughan et al., 2020). An aerial photo from 1994 showed that there was a forest with no apparent signs of hydrothermal activity (Vaughan et al., 2020). The hydrothermal area slowly expanded from 2000 until 2005, then started expanding more rapidly for the next 10 years before stabilizing at \sim 33,300 m² around 2015 (Vaughan et al., 2020). This expansion was primarily identified by measuring the size of the tree kill zone, which is likely caused by some combination of excessive heat in the soil, sulfur gasses, or excessively high CO₂ concentrations in the soil (Farrar et al., 1995; Jenkins et al., 2012; Vaughan et al., 2020). Infrared camera images showed an arc-shaped zone of high temperatures above 50°C; but outside this zone, the soil temperatures generally return to ambient levels within a few meters (Vaughan et al., 2020). Vaughan et al. (2020) characterized this as an acid-sulfate area, a type of hydrothermal zone typically associated with high CO₂ and sulfur gas emissions (Vaughan et al., 2020; Werner & Brantley, 2003). These inferred gas emissions were confirmed by the presence of steaming fumaroles and the smell of H₂S (Vaughan et al., 2020). The trees present in the hydrothermal zone and the area surrounding it have been exposed to significant quantities of volcanic gases in the last two decades, and Vaughan et al. (2020) recommended radiocarbon dendrochronology to determine the area's history of CO₂ emissions (Cook et al., 2001; Evans et al., 2010; Lewicki & Hilley, 2014; Vaughan et al., 2020). The TLTA is an ideal study site due to the decadal-scale growth of the affected area and the presence of undisturbed and otherwise virtually identical forests in close proximity. The progressive evolution of the area allows us to study plant responses to different types and intensities of volcanic activity, and the homogeneity of the species in the forest reduces the number of factors which could complicate interpretation.

2.2. Remote Sensing Methods

The NDVI is one of the most widely used vegetation indices for remote sensing. Chlorophyll reflects near-infrared (NIR) light and absorbs red light; therefore, areas which have high concentrations of chlorophyll will have higher NIR values and lower red values (Richardson, 1977). Healthy plants have higher chlorophyll concentrations than stressed plants, so measuring chlorophyll with NDVI provides information on the general health of plants in an area. NDVI is calculated using the red and NIR bands:

$$NDVI = \frac{NIR - red}{NIR + red}$$
(1)

This calculation yields unitless values which range from -1 to 1, with higher values indicating healthier vegetation. Dense green forest commonly has values between 0.6 and 1, whereas less healthy vegetation will have lower values. Bare rock typically has values around 0, and water will usually have negative values (Jones & Vaughan, 2010).

We used publicly available Landsat 5 and Landsat 8 image databases provided by the United States Geological Survey (USGS). Each image has a 30 m pixel size and was processed for surface reflectance by the USGS





Figure 1. Location of study site within Yellowstone National Park, Wyoming, USA. Park boundaries (black line) and the caldera rim (orange line) shown in top right. The satellite image shown is a Maxar (Vivid) image captured on 17 October 2022, at a resolution of 0.3 m and was sourced from the ESRI World Imagery basemap.

(Vermote et al., 2016). We removed any images which had clouds present in the study area and calculated NDVI values for the remaining 886 images. We also discarded any images where the first background site had an average NDVI value of less than 0.4, as that was the lowest value we observed during any of the primary growing season months (June–September) and it is the closest background area to the thermal area. We assume that any images which have a value lower than this are being affected by snow, wildfires, or other factors which could reduce differences between the background area and study site and make comparisons more prone to outliers and misinterpretation. Our methodology is designed to eliminate as many external factors that could be affecting the vegetation as possible so that we can focus exclusively on the effects of the hydrothermal activity. These constraints removed 701 images from the data set, leaving a total of 185 usable images (see Supporting Information S1 for full list of used images). We used ChatGPT to assist with writing and troubleshooting some of the Python code used to download the images and produce Figures 2–5. We reviewed all code taken from ChatGPT to ensure it was accurate and functioned as intended.

We chose the forest immediately north of the TLTA as the first background area for our study (Figure 1). The second and third background areas are in another similar forest, which is 2 km away from the thermal area. We drew the polygons for the background areas with the intent to avoid any clearings in the forest and to include as much contiguous forest as possible, without including any pixels that were near the edge, as these pixels may contain less dense tree cover or bare ground. The three background areas have areas of 41,700, 80,463, and 73,501 m². The TLTA and the first background site are less than 100 m apart and, prior to the expansion of the hydrothermal area in the early 2000s, were both part of one contiguous and relatively homogeneous forest. Diffuse volcanic carbon dioxide emissions enhance air concentrations of CO₂ within a small spatial area, typically returning to background values within a few tens of meters (Fernández et al., 1998; Lewicki et al., 2017; Marín et al., 2005; Rahilly & Fischer, 2021). Diffuse H₂O emissions probably dissipate similarly, although we are unaware of any studies which have specifically measured this effect in a volcanic or hydrothermal setting. Our first background area is more than 60 m from the thermal area, so there is very low probability of any significant elevated CO_2 exposure in the background area. Recent fieldwork by one of the authors (RRB) at the TLTA confirmed that the forests in the area consist almost entirely of lodgepole pine (Pinus cortata), the most common type of tree in Yellowstone National Park (Despain, 1990; Yellowstone Volcano Observatory, 2019). This background site is ideal because it is near the study site, has similar terrain, elevation, slope, and aspect, and both sites had the same tree species prior to the hydrothermal activity. The second and third background sites are \sim 2 km east of the TLTA and at similar elevations (\sim 2,530 m). These sites were chosen due to their proximity to



Geochemistry, Geophysics, Geosystems



Figure 2. (a) Array of images showing an aerial photo, (b) a satellite photo, and (c, d) Normalized Difference Vegetation Index images of the thermal area and the first background area in 1994 and 2022. (a) Is sourced from the National Aerial Photography Program, (b) is a MAXAR (Vivid) photo from the ESRI World Imagery Basemap, (c) is a Landsat 5 image, and (d) is a Landsat 8 image.

the TLTA and their relatively stable, continuous, and homogenous forest (Figure 1). Neither site has any apparent differences in the typical factors which affect plant health, such as soil type, sunlight exposure, and water availability. The nearest hydrothermal area to either of these background sites is more than 500 m distant, and thus has no effect on the vegetation there.

The differences in NDVI values between the hydrothermal and background areas varied considerably over the time period studied. We define these differences as Δ NDVI:

$$\Delta NDVI = NDVI_{background area} - NDVI_{thermal area}$$
(2)

We compared Δ NDVI values of the hydrothermal and background areas using several different statistics for each area, including mean, median, maximum (max), minimum (min), 25th percentile (Q1), and 75th percentile



Geochemistry, Geophysics, Geosystems



Figure 3. Annual precipitation and mean Normalized Difference Vegetation Index (NDVI) values for the three background areas and the thermal area from 1984 to 2022. Dots indicate mean NDVI values for individual dates, and the lines represent yearly averages of these values. The vertical red line indicates the date of the first detected vegetation stress in Vaughan et al. (2020).

(Q3). For each of these statistics, we calculated the statistic for each area and then compared the two values using Equation 2. For example, the Δ NDVI_{mean} for the first background area is the mean value of the thermal area subtracted from the mean value of the first background area. When these values are negative, the hydrothermal area has higher NDVI values than the background area, suggesting that the hydrothermal activity is promoting plant health by CO₂ fertilization or increased access to water. When these values are positive, the background area is healthier than the TLTA, suggesting that the vegetation in the TLTA is stressed by hydrothermal sulfur emissions, excessively high CO₂ concentrations, and/or high soil temperatures. Trends in these values are significant too; if the values are positive but decreasing, the thermal area still has lower NDVI values than the background area area, but the decreasing difference indicates that plant health in the thermal area is increasing relative to the background area are experiencing different effects than other parts. We assume that in the absence of hydrothermal activity, the hydrothermal area and the background areas should have nearly identical values for most or all of these metrics.

3. Results

From 1984 to 2000, NDVI values in the background areas and the TLTA typically ranged from 0.40 to 0.70 (Figure 3). Mean NDVI values in the TLTA and background areas decreased slightly in 1988, likely due to the wildfires that swept through Yellowstone that year. The quick rebound in NDVI values during the following year confirms that this area did not burn, but the vegetation was still affected in 1988 by the low precipitation and





Figure 4. (a) Δ NDVI metrics over time for the first background area, (b) the second background area, and (c) the third background area relative to the thermal area for the time period 1985–2022. The lines in this figure represent yearly average values. The vertical red line indicates the date of the first detected vegetation stress in Vaughan et al. (2020). Positive values indicate that the vegetation in the background area is healthier than that in the thermal area, and negative values indicate the opposite. See Equation 2 for explanation of Δ NDVI.

potentially also smoke from the fires. Precipitation in Yellowstone was relatively low from 1984 to 1998, averaging 506.1 mm yr⁻¹ (Figure 3) (Menne et al., 2012a, 2012b). From 1998 to 2023, average yearly precipitation nearly doubled to 892.8 mm yr⁻¹. The mean NDVI in the three background areas from 1998 to 2023 was 9.5%-12.7% higher than from 1984 to 1998, likely driven by the increased water availability from precipitation.

All Δ NDVI metrics for the three background areas range from 0.04 to 0.16 at the beginning of the study period, with most metrics near 0.1 (Figure 4). The metrics for each background area from 1985 to 2001 have their own yearly variation, but all three follow the same longer-term trend of decreasing to near 0 or slightly negative values by 2001. After 2001, all metrics start increasing, with $\Delta NDVI_{min}$ and $\Delta NDVI_{O1}$ increasing the most rapidly. The next major inflection point in the graph occurs in 2006, where the rest of the metrics start increasing much more rapidly and generally continue to increase for the remainder of the study period. The $\Delta NDVI_{max}$ was the last metric to begin increasing and increased relatively slowly until 2008. $\Delta NDVI_{mean}$, $\Delta NDVI_{median}$, and $\Delta NDVI_{Q3}$ typically followed nearly identical trends throughout the study period, and until 2003 the $\Delta NDVI_{01}$ behaved similarly. The $\Delta NDVI_{min}$ and $\Delta NDVI_{max}$ mostly followed the same broader trends but with more variation due to the inherent volatility of measuring only a single pixel as opposed to the other metrics which incorporate values from a greater number of pixels.

We also examined the spatial distribution of the pixels containing the highest and lowest NDVI values each year, with a particular focus before 2000, which was the previously recognized start of the expansion of the hydrothermal area. Theoretically, the lowest-value pixels should be trees negatively impacted by hydrothermal activity and could potentially indicate the epicenter of any hydrothermal activity that stressed vegetation. The location of the highest NDVI values could indicate where vegetation was positively impacted by hydrothermal activity, either by CO_2 fertilization or increased access to water. The image from each year which had the highest minimum NDVI value was chosen, which was typically in the middle of July but sometimes as late as the end of September and as early as late June. This approach mitigates any seasonal effects by ensuring that the image chosen is in the peak of the growing season and facilitates comparison between years. For each image, the pixels which fall in the highest and lowest bins in a 10-bin histogram were illustrated (see Figure 5 and Figure S1 in Supporting Information S1).

From 1985 to 1992, the lowest NDVI values were generally in the southernmost part of the hydrothermal area (Figure 5). From 1993 to 2001, the location of the lowest values switched to the northernmost part, except for 1996–1997, where it flipped back to the south. This northern area is the same zone where hydrothermal activity was first detected from satellite thermal infrared images in 2000 (Vaughan et al., 2020). Starting in 1999, one of the lowest value pixels was in the center of the hydrothermal area, which became the main epicenter of hydrothermal activity. This pixel consistently had one of the lowest values from 1999 onwards, consistent with images from Vaughan et al. (2020), which show a bare patch of dead trees emerging in the center of the hydrothermal area starting in 2002. The highest values in the

thermal area are typically on the western edge, although they move more to the center and eastern side from 1997 to 1998. In the first background area, the highest values are commonly on the eastern side, while the lowest values move around from year to year with no clear pattern (Figure S2 in Supporting Information S1).





Figure 5. Colored pixels indicate the location of the pixels with the values in the highest and lowest bins in a 10 bin histogram within the Tern Lake thermal area from selected images from 1991 to 2010. Each pixel is colored based on how many standard deviations the Normalized Difference Vegetation Index (NDVI) value of that pixel is from the mean NDVI value for that image (*z*-score). These four images were chosen because each is representative of several years of activity. A similar figure which includes 32 images from 1985 to 2018 is in Figure S1 in Supporting Information S1, as well as another figure S2 in Supporting Information S1).

4. Discussion

We identify three notable periods from 1984 to 2022, which represent different types of activities in the study area: 1984–1999, 1999–2006, and 2006–2022. The TLTA evolved rapidly over the duration of the study period, which provides an opportunity to analyze how NDVI changes in response to different types and intensities of hydrothermal activity. The changes in activity after 2000 are well documented by the visible expansion of the tree kill area and the development of thermal anomalies. We use the NDVI responses to interpret the NDVI variations observed prior to 2000. In the absence of hydrothermal activity, we expect background and thermal areas to have relatively similar values for these metrics.

4.1. Moderate Activity: 1984–1999

The first period (1984–1999) is characterized by a gradual decline in all Δ NDVI metrics from all background areas. This means that the NDVI of the TLTA was generally increasing relative to the background areas throughout this time period. We interpret this as moderate hydrothermal activity promoting growth throughout the thermal area due to increased access to water or

CO₂ fertilization. This interpretation also implies that hydrothermal activity at the TLTA started at least 15 years earlier than previously recognized, and that the thermal anomalies observed in 2000 actually represent a reawakening or intensification of activity at the site, as opposed to the onset of activity. This hypothesis is supported by the occurrence of a series of small earthquakes in the early 1980s within 5 km of the thermal area, which are commonly associated with movement of magmatic fluids (Greenfield et al., 2019; Roman & Cashman, 2006; U.S. Geological Survey, 2022). Some of the later expansion of the hydrothermal area is associated with an earthquake swarm in 2010, so this earlier earthquake swarm may also have signified an increase in hydrothermal activity, which then declined over the following decade (Vaughan et al., 2020). The Δ NDVI metrics for the first background area are relatively stable, but the Δ NDVI metrics for the other two background areas are more volatile. There are major drops in Δ NDVI in 1989 and 1992–1993 for both background 2 and 3, which are much less pronounced in the first background area. The first background area is closer to the thermal area than the other two, so this may indicate that there are some climatic conditions that were more strongly affecting the second and third background areas than the first background and thermal areas. Since all three background areas follow the same trends, with varying degrees of intensity, this could also suggest that there were short-term changes in hydrothermal activity in 1989 and 1992–1993, which promoted plant health in the TLTA.

From 1985 to 1991, the lowest NDVI values were located at the southern tip of the hydrothermal area (Figure 5), indicating that hydrothermal activity during this time was probably centered there instead of the northern and central locations where the post-2000 activity first became visible (Vaughan et al., 2020). The location of the lowest value pixel in the TLTA also started migrating in 1992, moving from the southern tip to the center and then to the northern tip of the area in 1993. Except for 1996–1997, the lowest NDVI values are consistently at the northern edge of the thermal area during this time, which is where one of the first infrared temperature anomalies appears in 2000 (Vaughan et al., 2020). This anomaly was first detected in an image from 2000, whereas the lowest NDVI values began appearing there in 1993, 7 years earlier. This may indicate that vegetation in this area was already beginning to be stressed by hydrothermal activity in the early 1990s, several years before it became apparent in thermal infrared or visible images. Fracture pathways through which hydrothermal gases and fluids travel can seal over time due to mineral deposition (Boudon et al., 1998; Edmonds et al., 2003; Stix et al., 1997). The switch of the epicenter of activity from the south to the north that we observe in 1992–1993 may reflect the sealing of the primary transport pathway that fueled activity in the south. We suggest that vegetation can respond to changes in hydrothermal activity long before it becomes apparent from thermal or visible imagery, highlighting its potential for early detection of changes in volcanic activity.

4.2. Intensification of Activity: 1999-2006

The second period is characterized by stronger magmatic CO_2 and/or H_2O fertilization and the beginning of the expansion of the tree kill area. In 1999, the $\Delta NDVI$ metrics for background areas 1 and 3 start decreasing more

8 of 12

rapidly and some of the lowest NDVI value pixels start appearing in the center of the TLTA. The location of these pixels becomes the center of the tree kill area in subsequent years and is the part of the TLTA with the highest measured temperatures (Vaughan et al., 2020). This is another example of vegetation responding to hydrothermal activity before it is detectable with other remote sensing methods. By 2001, the $\Delta NDVI_{mean}$ metrics for all three background areas dropped below 0, the only year in the study period when this occurred. This is the year after the earliest signs of hydrothermal activity were detected in this area, and it strongly suggests that CO₂ fertilization and/or increased access to magmatic water drove a relative increase in NDVI in the thermal area compared to previous years. After 2001, the $\Delta NDVI_{01}$ and $\Delta NDVI_{min}$ started increasing for all background areas, reflecting the initiation of major vegetation stress in small parts of the thermal area. The other metrics are less consistent, and most of them stay relatively flat until 2006. This indicates that only a small portion of the thermal area in the center and the northern edge had sufficiently intense heat or sulfur emissions to stress plants during this time, while most of the rest of the thermal area was largely unaffected by these negative impacts. This corresponds well with the visible and satellite imagery, which show that thermal output and the size of the tree kill area increased slowly until 2006. From 2002 to 2008, the lowest NDVI values in the thermal area are located exclusively in the center of the TLTA. This indicates that the epicenter of hydrothermal activity in the TLTA has moved from the northern edge to the middle, where it has remained until the present day.

4.3. Activity Peaks: 2006-2022

All Δ NDVI values started rapidly increasing in 2006, reflecting the "rapid growth" period described by Vaughan et al. (2020) and indicating that intensifying hydrothermal activity was severely stressing the vegetation in the thermal area. NDVI values in the thermal area continue decreasing rapidly until 2011, after which they continued decreasing but at a slower rate (Figure 4d). By 2011, the hydrothermal area was nearly barren, so the variations in the Δ NDVI metrics at this point are more likely related to fluctuations in the background areas' NDVI values rather than actual changes in hydrothermal activity. Vaughan et al. (2020) noticed significant new tree growth within the bounds of the tree kill area during their field work in August 2019, indicating that hydrothermal activity may be declining in recent years. The trees present there are still very young and too small to have a major effect on NDVI, but if hydrothermal activity does not re-intensify, we would expect NDVI in the thermal area to continue to rise in coming years and for some of the Δ NDVI metrics to start to decrease as these new trees continue to grow.

4.4. Future Directions

This approach holds significant promise for future work, although there are some important issues to consider. This project focused on a relatively small area, but our general approach should be applicable to much larger areas as well. The broad flanks of stratovolcanoes consistently emit volatiles and are commonly forested, making them excellent targets for future projects. Volcanoes with well-developed, relatively homogenous, and healthy forests are the ideal targets. Plants that are under significant stress from normal environmental factors such as droughts, wildfires, or high air temperature are less useful for this approach because negative impacts from volcanic sulfur emissions or high soil temperatures will be difficult to disentangle from other non-volcanic stresses. This approach will be easiest to implement in areas similar to the TLTA, which only have one dominant tree species, as the presence of multiple species can influence NDVI values. The spatial distribution of different tree species could lead to certain pixels having lower or higher NDVI values which are unrelated to volcanic or hydrothermal influences, and this issue must be considered in more complicated forest environments. Most volcanoes in arid climates or at high elevations are unlikely to be viable for this approach due to sparse vegetation, which does limit its utility in some major volcanic regions such as the central Andes. Some volcanoes are also cloudy for large parts of the year, which can make it difficult to acquire sufficient cloud-free images to conduct a robust study. Additionally, the single pixel analysis of the spatial variability of high and low NDVI values highlights the need for higher resolution imagery.

In many cases, the ecological effects of volcanic activity are likely subtle, and therefore easy to overlook for human observers. Major large-scale NDVI anomalies like the ones observed at Mt. Etna and Nyiragongo are easily noticeable without deeper analysis (Houlié et al., 2006), but these two examples probably represent a relatively rare phenomenon. These major NDVI anomalies are most likely caused by increases in soil water sourced from degassing magma (Seiler et al., 2021), but more work needs to be done to explore how the magnitude of

these effects may vary under different conditions. Increased CO₂ also increases water use efficiency in plants, which may enhance the positive effects of access to magmatically sourced water (Drake et al., 1997). We suspect that most forested, active volcanoes will have certain areas with subtle effects on the local vegetation, which may require large amounts of data and high-level statistical or machine learning analysis for detection. We suggest that these more subtle effects are the most crucial for future studies, as they hold the most potential for widespread use at volcanoes around the world. There is an extensive body of ecological research focused on how ecosystems react to elevated CO₂ concentrations, which could provide a number of different approaches for expanding our techniques (see reviews by Ainsworth and Long (2005) and Saban et al. (2019)). This study focused on NDVI, a relatively simple metric for understanding vegetation health, but future studies could explore specific effects which have been observed in experimental and natural settings, such as increased biomass, increased tree height, increased specific leaf area, increased leaf starch content, and decreased stomatal conductance (Ainsworth & Long, 2005; Saban et al., 2019). A further goal is to demonstrate the potential for gathering ecosystem-scale data on plant responses to CO₂ to provide new insight into how forest ecosystems respond to rising levels of atmospheric CO₂. Our understanding of how elevated concentrations of CO₂ affect ecosystems has improved considerably in recent years, but there is still a notable lack of field-based data regarding how ecosystems in non-temperate environments react to these conditions (Walker et al., 2021). Future studies of ecosystem responses to volcanic CO₂ emissions could both benefit from and contribute to this body of knowledge, and we strongly encourage collaboration with ecologists to take advantage of this mutual interest.

Ground-based validation will be crucial for confirming the results of our study. Tree ring studies following the model of Seiler et al. (2021) could confirm our findings and help distinguish whether volcanic CO₂, H₂O, or some other previously unconsidered factor is responsible for the trends observed. In situ measurements of vegetation traits and functional responses in degassing areas could help identify the strongest plant responses to different types of volcanic degassing. Future studies could then focus on using satellites which are best suited to detecting these responses. We have refrained from providing quantitative estimates of volatile emissions in this paper, but future projects may be able to develop models with ground truthing data which could make yearly estimates of diffuse volcanic volatile emissions in an area of interest possible. This would require ground-based measurements of diffuse volcanic degassing and nearby vegetation, but could be an effective way to scale up degassing measurements from a small area to an entire volcanic edifice, or to identify areas of degassing to target for future ground-based degassing measurements. The TLTA is a relatively small area, so it should be possible to acquire ground truthing data that would be representative of the entire area. If a future project scaled up this approach to a larger area, such as a large volcanic edifice, then ground truthing select portions of the area of interest could allow for quantification of volatile emissions across the entire edifice. Combining ground-based plant and gas emission measurements with our satellite approach could yield a much more comprehensive understanding of volcanic volatile emissions than either approach individually.

5. Conclusions

Hydrothermal activity at the TLTA was first recognized from satellite detection of infrared temperature anomalies and visual examination of the expansion of the tree kill zone (Vaughan et al., 2020). Our further investigation of the satellite image record focused on measuring plant responses using NDVI demonstrates that hydrothermal activity in this area may have begun significantly earlier than previously recognized, and that the activity observed in the early 2000s is likely an intensification of ongoing hydrothermal activity. The declining Δ NDVI values in 1984–2001 support the hypothesis that there was hydrothermal activity which was promoting plant health in the TLTA during this period. NDVI is higher in the thermal area than the background areas for all measured statistics in 2001, which is shortly after Vaughan et al. (2020) recognized the start of hydrothermal activity, suggesting that increased hydrothermal activity in 2001 released large quantities of CO₂ and/or H₂O, which promoted plant health. The later expansion of the hydrothermal activity is well tracked by NDVI, and the intensification of activity in 2006 is reflected by large increases in all Δ NDVI metrics. We demonstrate the value of NDVI as a tool for examining past and current volcanic and hydrothermal activity.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

All Landsat 5 and 8 images used in this article are publicly available through the USGS EarthExplorer interface: https://earthexplorer.usgs.gov/. All calculated statistics for the thermal and background areas as well as a complete list of images used are available at Bogue (2023).

References

- Ainsworth, E. A., & Long, S. P. (2005). What have we learned from 15 years of free-air CO₂ enrichment (FACE)? A meta-analytic review of the responses of photosynthesis, canopy properties and plant production to rising CO₂. *New Phytologist*, *165*(2), 351–372. https://doi.org/10.1111/j.1469-8137.2004.01224.x
- Bogue, R. R. (2023). Tern Lake thermal area Landsat NDVI data [Dataset]. Borealis. https://doi.org/10.5683/SP3/K1JWSO
- Bogue, R. R., Schwandner, F. M., Fisher, J. B., Pavlick, R., Magney, T. S., Famiglietti, C. A., et al. (2018). Plant responses to volcanically-elevated CO₂ in two Costa Rican forests. *Biogeosciences Discussions*, 1–27. https://doi.org/10.5194/bg-2018-100
- Bottinga, Y., & Javoy, M. (1990). MORB degassing: Bubble growth and ascent. *Chemical Geology*, 81(4), 255–270. https://doi.org/10.1016/0009-2541(90)90050-H
- Boudon, G., Villemant, B., Komorowski, J.-C., Ildefonse, P., & Semet, M. P. (1998). The hydrothermal system at Soufriere Hills Volcano, Montserrat (West Indies): Characterization and role in the on-going eruption. *Geophysical Research Letters*, 25(19), 3693–3696. https://doi. org/10.1029/98GL00985
- Bull, K. F., Cameron, C. E., Coombs, M. L., Diefenbach, A. K., Lopez, T., McNutt, S. R., et al. (2011). The 2009 eruption of Redoubt volcano, Alaska (Report of investigations 2011–5). Alaska Division of Geological & Geophysical Surveys. https://doi.org/10.14509/23123
- Burton, M. R., Sawyer, G. M., & Granieri, D. (2013). Deep carbon emissions from volcanoes. *Reviews in Mineralogy and Geochemistry*, 75(1), 323–354. https://doi.org/10.2138/rmg.2013.75.11
- Carn, S. A., Fioletov, V. E., McLinden, C. A., Li, C., & Krotkov, N. A. (2017). A decade of global volcanic SO₂ emissions measured from space. *Scientific Reports*, 7(1), 44095. https://doi.org/10.1038/srep44095
- Cawse-Nicholson, K., Fisher, J. B., Famiglietti, C. A., Braverman, A., Schwandner, F. M., Lewicki, J. L., et al. (2018). Ecosystem responses to elevated CO₂ using airborne remote sensing at Mammoth Mountain, California. *Biogeosciences*, 15(24), 7403–7418. https://doi.org/10.5194/ bg-2018-73
- Chiodini, G., Granieri, D., Avino, R., Caliro, S., Costa, A., & Werner, C. (2005). Carbon dioxide diffuse degassing and estimation of heat release from volcanic and hydrothermal systems. *Journal of Geophysical Research*, 110(B8), B08204. https://doi.org/10.1029/2004JB003542
- Christiansen, R. L. (2001). The Quaternary and Pliocene Yellowstone Plateau volcanic field of Wyoming, Idaho, and Montana (p. 145). U.S. Geological Survey Professional Paper. 729-G. https://doi.org/10.3133/pp729G
- Committee on Improving Understanding of Volcanic Eruptions, Committee on Seismology and Geodynamics, Board on Earth Sciences, Division on Earth and Life Studies, & National Academies of Sciences, Engineering, and Medicine. (2017). Volcanic eruptions and their Repose, unrest, precursors, and timing (p. 24650). National Academies Press. https://doi.org/10.17226/24650
- Cook, A. C., Hainsworth, L. J., Sorey, M. L., Evans, W. C., & Southon, J. R. (2001). Radiocarbon studies of plant leaves and tree rings from Mammoth Mountain, CA: A long-term record of magmatic CO₂ release. *Chemical Geology*, 177(1–2), 117–131. https://doi.org/10.1016/ S0009-2541(00)00386-7
- Cox, P., Pearson, D., B Booth, B., Friedlingstein, P., Huntingford, C., Jones, C., & M Luke, C. (2013). Sensitivity of tropical carbon to climate change constrained by carbon dioxide variability. *Nature*, 494(7437), 341–344. https://doi.org/10.1038/nature11882
- D'Arcy, F., Boucher, É., Moor, J. M. D., Hélie, J.-F., Piggott, R., & Stix, J. (2019). Carbon and sulfur isotopes in tree rings as a proxy for volcanic degassing. *Geology*, 47(9), 825–828. https://doi.org/10.1130/G46323.1
- Despain, D. G. (1990). Yellowstone vegetation: Consequences of environment and history in a natural setting. Roberts Rinehart Publishers.
- Drake, B. G., Gonzàlez-Meler, M. A., & Long, S. P. (1997). More efficient plants: A consequence of rising atmospheric CO₂? Annual Review of Plant Biology, 48(1), 609–639. https://doi.org/10.1146/annurev.arplant.48.1.609
- Edmonds, M., Oppenheimer, C., Pyle, D. M., Herd, R. A., & Thompson, G. (2003). SO₂ emissions from Soufrière Hills Volcano and their relationship to conduit permeability, hydrothermal interaction and degassing regime. *Journal of Volcanology and Geothermal Research*, 124(1), 23–43. https://doi.org/10.1016/S0377-0273(03)00041-6
- Evans, W. C., Bergfeld, D., McGeehin, J. P., King, J. C., & Heasler, H. (2010). Tree-ring ¹⁴C links seismic swarm to CO₂ spike at Yellowstone, USA. *Geology*, 38(12), 1075–1078. https://doi.org/10.1130/g31345.1
- Farrar, C. D., Sorey, M. L., Evans, W. C., Howle, J. F., Kerr, B. D., Kennedy, B. M., et al. (1995). Forest-killing diffuse CO₂ emission at Mammoth Mountain as a sign of magmatic unrest. *Nature*, 376(6542), 675–678. https://doi.org/10.1038/376675a0
- Fernández, M. D., Pieters, A., Donoso, C., Tezara, W., Azkue, M., Herrera, C., et al. (1998). Effects of a natural source of very high CO₂ concentration on the leaf gas exchange, xylem water potential and stomatal characteristics of plants of Spatiphylum cannifolium and Bauhinia multinervia. *New Phytologist*, 138(4), 689–697. https://doi.org/10.1046/j.1469-8137.1998.00161.x
- Finizola, A., Revil, A., Rizzo, E., Piscitelli, S., Ricci, T., Morin, J., et al. (2006). Hydrogeological insights at Stromboli volcano (Italy) from geoelectrical, temperature, and CO₂ soil degassing investigations. *Geophysical Research Letters*, 33(17), L17304. https://doi.org/10.1029/2006GL026842
- Fischer, T. P. (2008). Fluxes of volatiles (H₂O, CO₂, N₂, Cl, F) from arc volcanoes. *Geochemical Journal*, 42(1), 21–38. https://doi.org/10.2343/geochemj.42.21
- Friedlingstein, P., Meinshausen, M., Arora, V. K., Jones, C. D., Anav, A., Liddicoat, S. K., & Knutti, R. (2014). Uncertainties in CMIP5 climate projections due to carbon cycle feedbacks. *Journal of Climate*, 27(2), 511–526. https://doi.org/10.1175/JCLI-D-12-00579.1
- Gamon, J. A., Field, C. B., Goulden, M. L., Griffin, K. L., Hartley, A. E., Joel, G., et al. (1995). Relationships between NDVI, canopy structure, and photosynthesis in three californian vegetation types. *Ecological Applications*, 5(1), 28–41. https://doi.org/10.2307/1942049
- Gardner, J. E., Hilton, M., & Carroll, M. R. (2000). Bubble growth in highly viscous silicate melts during continuous decompression from high pressure. *Geochimica et Cosmochimica Acta*, 64(8), 1473–1483. https://doi.org/10.1016/S0016-7037(99)00436-6
- Girona, T., Costa, F., Taisne, B., Aggangan, B., & Ildefonso, S. (2015). Fractal degassing from Erebus and Mayon volcanoes revealed by a new method to monitor H₂O emission cycles. *Journal of Geophysical Research: Solid Earth*, *120*(5), 2988–3002. https://doi.org/10.1002/2014JB011797
- Girona, T., Realmuto, V., & Lundgren, P. (2021). Large-scale thermal unrest of volcanoes for years prior to eruption. *Nature Geoscience*, *14*(4), 238–241. https://doi.org/10.1038/s41561-021-00705-4
- Global Volcanism Program. (2014). Report on Fogo (Cape Verde). Bulletin of the Global Volcanism Network, 39(11). https://doi.org/10.5479/ si.GVP.BGVN201411-384010

Acknowledgments

We would first like to thank two anonymous reviewers for providing detailed and constructive reviews of this paper. Their suggestions greatly improved the quality of the manuscript, and we appreciate them taking the time to review this paper. We would also like to thank Dr. Marie Edmonds for handling editorial duties for this manuscript. We would like to thank Dr. Florian Schwandner for many productive discussions regarding the interactions between volcanic gases and ecosystems. We would also like to acknowledge the USGS for providing the Landsat 5 and 8 images used in this study. Robert Bogue was supported by a Vanier Canada Graduate Scholarship (#175782) during this project. Dr. John Stix and Dr. Peter Douglas were supported by NSERC Discovery Grants.

- Greenfield, T., Keir, D., Kendall, J.-M., & Ayele, A. (2019). Low-frequency earthquakes beneath Tullu Moye volcano, Ethiopia, reveal fluid pulses from shallow magma chamber. *Earth and Planetary Science Letters*, 526, 115782. https://doi.org/10.1016/j.epsl.2019.115782
- Houlié, N., Komorowski, J. C., de Michele, M., Kasereka, M., & Ciraba, H. (2006). Early detection of eruptive dykes revealed by normalized difference vegetation index (NDVI) on Mt. Etna and Mt. Nyiragongo. *Earth and Planetary Science Letters*, 246(3), 231–240. https://doi. org/10.1016/j.epsl.2006.03.039
- Jenkins, M. W., Krofcheck, D. J., Teasdale, R., Houpis, J., & Pushnik, J. (2012). Exploring the edge of a natural disaster. Open Journal of Ecology, 02(04), 222–232. https://doi.org/10.4236/oje.2012.24026
- Jervis, D., McKeever, J., Durak, B. O. A., Sloan, J. J., Gains, D., Varon, D. J., et al. (2021). The GHGSat-D imaging spectrometer. Atmospheric Measurement Techniques, 14(3), 2127–2140. https://doi.org/10.5194/amt-14-2127-2021
- Jones, H. G., & Vaughan, R. A. (2010). Remote sensing of vegetation: Principles, techniques, and applications. OUP Oxford.
- Kern, C., Aiuppa, A., & de Moor, J. M. (2022). A golden era for volcanic gas geochemistry? *Bulletin of Volcanology*, 84(5), 43. https://doi. org/10.1007/s00445-022-01556-6
- Kern, C., Masias, P., Apaza, F., Reath, K. A., & Platt, U. (2017). Remote measurement of high preeruptive water vapor emissions at Sabancaya volcano by passive differential optical absorption spectroscopy. *Journal of Geophysical Research: Solid Earth*, 122(5), 3540–3564. https:// doi.org/10.1002/2017JB014020
- Lewicki, J. L., & Hilley, G. E. (2014). Multi-scale observations of the variability of magmatic CO₂ emissions, Mammoth Mountain, CA, USA. Journal of Volcanology and Geothermal Research, 284, 1–15. https://doi.org/10.1016/j.jvolgeores.2014.07.011
- Lewicki, J. L., Kelly, P. J., Bergfeld, D., Vaughan, R. G., & Lowenstern, J. B. (2017). Monitoring gas and heat emissions at Norris Geyser Basin, Yellowstone National Park, USA based on a combined eddy covariance and Multi-GAS approach. *Journal of Volcanology and Geothermal Research*, 347, 312–326. https://doi.org/10.1016/j.jvolgeores.2017.10.001
- Marín, O., Rengifo, E., Herrera, A., & Tezara, W. (2005). Seasonal changes in water relations, photosynthesis and leaf anatomy of two species growing along a natural CO₂ gradient. *Interciencia*, 30(1), 33–38.
- Menne, M. J., Durre, I., Korzeniewski, B., McNeil, S., Thomas, K., Yin, X., et al. (2012a). Global historical climatology network—Daily (GHCN-daily), version 3.29. NOAA National Climatic Data Center. https://doi.org/10.7289/V5D21VHZ
- Menne, M. J., Durre, I., Vose, R. S., Gleason, B. E., & Houston, T. G. (2012b). An overview of the global historical climatology network-daily database. Journal of Atmospheric and Oceanic Technology, 29(7), 897–910. https://doi.org/10.1175/JTECH-D-11-00103.1
- Nuccio, P. M., & Paonita, A. (2001). Magmatic degassing of multicomponent vapors and assessment of magma depth: Application to Vulcano Island (Italy). *Earth and Planetary Science Letters*, 193(3), 467–481. https://doi.org/10.1016/S0012-821X(01)00512-X
- Proussevitch, A. A., & Sahagian, D. L. (1998). Dynamics and energetics of bubble growth in magmas: Analytical formulation and numerical modeling. *Journal of Geophysical Research*, 103(B8), 18223–18251. https://doi.org/10.1029/98JB00906
- Rahilly, K. E., & Fischer, T. P. (2021). Total diffuse CO₂ flux from Yellowstone caldera incorporating high CO₂ emissions from cold degassing sites. Journal of Volcanology and Geothermal Research, 419, 107383. https://doi.org/10.1016/j.jvolgeores.2021.107383
- Revil, A., Finizola, A., Ricci, T., Delcher, E., Peltier, A., Barde-Cabusson, S., et al. (2011). Hydrogeology of Stromboli Volcano, Aeolian Islands (Italy) from the interpretation of resistivity tomograms, self-potential, soil temperature and soil CO₂ concentration measurements. *Geophysical Journal International*, 186(3), 1078–1094. https://doi.org/10.1111/j.1365-246X.2011.05112.x
- Richardson, A. J. (1977). Distinguishing vegetation from soil background information. Photogrammetric Engineering & Remote Sensing, 43(12), 1541–1552.
- Roman, D. C., & Cashman, K. V. (2006). The origin of volcano-tectonic earthquake swarms. *Geology*, 34(6), 457–460. https://doi.org/10.1130/G22269.1
 Saban, J. M., Chapman, M. A., & Taylor, G. (2019). FACE facts hold for multiple generations; Evidence from natural CO₂ springs. *Global Change Biology*, 25(1), 1–11. https://doi.org/10.1111/gcb.14437
- Schwandner, F. M., Gunson, M. R., Miller, C. E., Carn, S. A., Eldering, A., Krings, T., et al. (2017). Spaceborne detection of localized carbon dioxide sources. *Science*, 358(6360), eeam5782. https://doi.org/10.1126/science.aam5782
- Seiler, R., Hajdas, I., Saurer, M., Houlié, N., D'Arrigo, R., Kirchner, J. W., & Cherubini, P. (2021). Tree-ring stable isotopes and radiocarbon reveal pre- and post-eruption effects of volcanic processes on trees on Mt. Etna (Sicily, Italy). *Ecohydrology*, 14(8), e2340. https://doi. org/10.1002/eco.2340
- Seiler, R., Kirchner, J. W., Krusic, P. J., Tognetti, R., Houlié, N., Andronico, D., et al. (2017). Insensitivity of tree-ring growth to temperature and precipitation sharpens the puzzle of enhanced pre-eruption NDVI on Mt. Etna (Italy). PLoS One, 12(1), e0169297. https://doi.org/10.1371/ journal.pone.0169297
- Shinohara, H. (2008). Excess degassing from volcanoes and its role on eruptive and intrusive activity. *Reviews of Geophysics*, 46(4), RG4005. https://doi.org/10.1029/2007RG000244
- Sparks, R. S. J., Barclay, J., Jaupart, C., Mader, H. M., & Phillips, J. C. (2018). Chapter 11a. Physical Aspects of Magma Degassing I. Experimental and theoretical constraints on vesiculation. In Volatiles in magma (pp. 413–446). De Gruyter. https://doi.org/10.1515/9781501509674-017
- Stix, J., Torres, R. C., Narváez, M. L., Cortés J, G. P., Raigosa, J. A., Gómez, M. D., & Castonguay, R. (1997). A model of vulcanian eruptions at Galeras Volcano, Colombia. *Journal of Volcanology and Geothermal Research*, 77(1), 285–303. https://doi.org/10.1016/ S0377-0273(96)00100-X
- Symonds, R. B., Gerlach, T. M., & Reed, M. H. (2001). Magmatic gas scrubbing: Implications for volcano monitoring. Journal of Volcanology and Geothermal Research, 108(1), 303–341. https://doi.org/10.1016/S0377-0273(00)00292-4
- U.S. Geological Survey. (2022). Earthquake hazards program, 2022, advanced National Seismic system (ANSS) comprehensive catalog of earthquake events and products: Various. Earthquake Hazards Program, 2022, Advanced National Seismic System (ANSS) Comprehensive Catalog of Earthquake Events and Products: Various. https://doi.org/10.5066/F7MS3QZH
- Vaughan, R. G., Hungerford, J. D. G., & Keller, W. (2020). A newly emerging thermal area in Yellowstone. Frontiers in Earth Science, 8, 204. https://doi.org/10.3389/feart.2020.00204
- Vermote, E., Justice, C., Claverie, M., & Franch, B. (2016). Preliminary analysis of the performance of the Landsat 8/OLI land surface reflectance product. *Remote Sensing of Environment*, 185, 46–56. https://doi.org/10.1016/j.rse.2016.04.008
- Walker, A. P., De Kauwe, M. G., Bastos, A., Belmecheri, S., Georgiou, K., Keeling, R. F., et al. (2021). Integrating the evidence for a terrestrial carbon sink caused by increasing atmospheric CO₂. New Phytologist, 229(5), 2413–2445. https://doi.org/10.1111/nph.16866
- Werner, C., & Brantley, S. (2003). CO₂ emissions from the Yellowstone volcanic system. *Geochemistry, Geophysics, Geosystems*, 4(7), 1061. https://doi.org/10.1029/2002GC000473
- Yellowstone Volcano Observatory. (2019). Yellowstone's newest thermal area: An up-close and personal visit!. Retrieved from https://www.usgs. gov/observatories/yvo/news/yellowstones-newest-thermal-area-close-and-personal-visit
- Zimmer, M., & Erzinger, J. (2003). Continuous H₂O, CO₂, ²²²Rn and temperature measurements on Merapi volcano, Indonesia. *Journal of Volcanology and Geothermal Research*, 125(1), 25–38. https://doi.org/10.1016/S0377-0273(03)00087-8